Author's response to reviews

Title: The Effect of Posterior Subtenon Methylprednisolone Acetate in the Refractory Diabetic Macular Edema: a prospective nonrandomized interventional case series

Authors:

Alireza Javadzadeh (javadzadehalireza@yahoo.com)

Version: 2 Date: 12 January 2006

Author's response to reviews:

Approval for the study was obtained from the ethical committee of Tabriz University of Medical Sciences which is in compliance with the Helsinki Declaration. All patients received a thorough explanation of the study design and aims, and were provided with written informed consent.